Author's response to reviews

Title: 'The girl with her period is the one to hang her head' Reflections on menstrual management among schoolgirls in rural Kenya

Authors:

  Shannon A McMahon (smcmahon@jhsph.edu)
  Peter J Winch (pwinch@jhsph.edu)
  Bethany Caruso (bcaruso@emory.edu)
  Alfredo Obure (afxobure@gmail.com)
  Emily A Ogutu (emily.awino@yahoo.com)
  Imelda Akinyi (imelda.akinyi@gmail.com)
  Richard Rheingans (rrheing@emory.edu)

Version: 4 Date: 26 May 2011

Author's response to reviews: see over
Dear Ms. Neilan:

We wanted to thank you and the reviewers for reviewing our manuscript. We are pleased that the manuscript was positively reviewed.

Because we cannot provide written informed consent for the imaged included as supplementary file 1, we ask that the image be removed. As requested, we have conducted a final review of the document and made minor formatting changes. We used track changes to highlight this.

Please do not hesitate to contact me with any questions.

Sincerely,

Peter Winch MD MPH
Professor and Director, Social and Behavioral Interventions Program
Associate Chair, Department of International Health